Author’s response to reviews

Title: Investigating the multi-target pharmacological mechanism of danhong injection acting on unstable angina by combined network pharmacology and molecular docking

Authors:
Siyu Guo (gsiyu1995@163.com)
Jia-Rui Wu (exogamy@163.com)
Wei Zhou (weizhou19940530@163.com)
Xinkui Liu (lxkchuige@163.com)
Jingyuan Zhang (lindajyz@163.com)
Shanshan Jia (cristielove@163.com)
Ziqi Meng (mengziq_2525jy@126.com)
Shuyu Liu (liushuxixi@163.com)
Mengwei Ni (nmh15764346161@163.com)
Yingying Liu (liuyingyinginn97@163.com)

Version: 1 Date: 12 Dec 2019

Dear Editor,

Thank you very much for your critical comments on our manuscript “Investigating the Multi-Target Pharmacological Mechanism of Danhong Injection Acting on Unstable Angina by Combined Network Pharmacology and Molecular Docking ” (ID: BCAM-D-19-02015). We have carefully corrected our submission, and amended manuscript were resubmitted. A point to point response was answered below:

Response : Thank you for your suggestion. We don't change the authors or the order of the authors.

Editor's comments:
Response: 1. Thank you for your suggestion. We have removed or amended the manuscript sentences the manuscript you mentioned.

2. Thank you for your suggestion. We have added the names of all the ingredients of Danhong Injection and supplementary materials have uploaded.

Editorial Policies

Response: Thank you for your suggestion. We have already added and amended the declarations sections.

We appreciate for Editors and Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions. If you have any questions, please don’t hesitate to contact us.

Thank you and best regards.

Yours sincerely,

Jiarui Wu

Corresponding author:
Name: Jiarui Wu
E-mail: exogamy@163.com